Author's response to reviews

Title: Use of the GenoType(R) MTBDRplus assay to assess drug resistance of Mycobacterium tuberculosis isolates from patients in rural Uganda

Authors:

Joel Bazira (jap2k2001@yahoo.co.uk)
Benon B Asiimwe (basiimwe@chs.mak.ac.ug)
Moses Joloba (moses.joloba@case.edu)
Fred Bwanga (fbwanga@chs.mak.ac.ug)
Mecky I Matee (mmateemi@yahoo.com)

Version: 6 Date: 6 August 2010

Author's response to reviews: see over
EDITOR’S COMMENTS

Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is 'First name Initial Last name' (e.g. Joe F Bloggs)

Response: We have listed the authors as advised.

Affiliations: Each affiliation should be written in full, in the following format: Department, Institute, (Street), City, (Postal code), Country. This may mean repeating institutional addresses if the authors are from different departments within an institution

Response: This has been done. Full addresses are now shown.

Tables: you have uploaded the tables as figure files. Please remove them from the submission system and include the tables within the text file of the manuscript following the references.

Response: Done

The tables should not be embedded images or drawings and should be formatted from cells, not tabbed text.

Response: Done

Please also move the table title to above the table and the legend to below the table, within the text.

Response: This has been done

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Email address: please include email addresses for all the authors on the title page, using the following format:
AB: abcd@institution.ac.uk
EF: efg@generic.co.uk
IJ: IJKL@corporation.com

Response: This has been done as advised.
Introduction: please rename this section 'Background'

Response: This has been done

Coloured text: please ensure that all text is black.

Response: All text is now in black and white

Tables- please remove the visible vertical lines from your tables

Response: We have tried but not successful, kindly assist here. Thank you.

Tables: please note that we are unable to correctly display merged cells where the merged cell crosses rows: please re-layout your table without these merged elements

Response: We have formatted our tables in such a way as to best relay the information to our readers. We sincerely hope that your system will be able to display the tables. However, we will be happy if you assist where it is really needed. Thank you.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

Response: We have read through the manuscript. We are happy with the current form. All authors have read and approved the current version.